

Author's response to reviews

Title: Operational challenges in managing Isoniazid Preventive Therapy in child contacts: A high-burden setting perspective

Authors:

Susan S van Wyk (susanvanwyk@sun.ac.za)
Anthony J Reid (tony.REID@brussels.msf.org)
Anna M Mandalakas (amm13@case.edu)
Donald A Enarson (denarson@theunion.org)
Nulda Beyers (nb@sun.ac.za)
Julie Morrison (juliem@sun.ac.za)
Anneke C Hesseling (annekeh@sun.ac.za)

Version: 4 **Date:** 6 July 2011

Author's response to reviews:

Major revisions

Tables: The table that was uploaded as a figure file has been removed from the submission system and the table has been included within the text file of the manuscript following the references.

Minor revisions

Affiliations: The country in affiliation 2 has been included.

Tracking: All tracking and coloured text has been removed from the manuscript.